Author's response to reviews

Title: Lamotrigine versus inert placebo in the treatment of borderline personality disorder: study protocol for a randomized controlled trial and economic evaluation.

Authors:

Mike J Crawford (m.crawford@imperial.ac.uk)
Rahil Sanatinia (r.sanatinia@imperial.ac.uk)
Barbara Barrett (barbara.barrett@kcl.ac.uk)
Sarah Byford (s.byford@kcl.ac.uk)
Gillian Cunningham (gillian.cunningham1@nhs.net)
Kavi Gakhal (k.gakhal@imperial.ac.uk)
Geoff Law (Geoff.Lawrence-Smith@oxleas.nhs.uk)
Verity Leeson (v.leeson@imperial.ac.uk)
Fenella Lemonsky (flemonsky@sky.com)
Georgia Lykomitrou (georgia.lykomitrou@nottshc.nhs.uk)
Alan Montgomery (Alan.Montgomery@nottingham.ac.uk)
Richard Morriss (Richard.Morriss@nottingham.ac.uk)
Carol Paton (c.paton@imperial.ac.uk)
Wei Tan (Wei.Tan@nottingham.ac.uk)
Peter Tyrer (p.tyrer@imperial.ac.uk)
Joseph G Reilly (j.g.reilly@durham.ac.uk)

Version: 5 Date: 26 June 2015

Author's response to reviews:

We have made the style changes requested by Hannah Lippman on 23rd June. I have followed the instructions on Contribution of the authors, but this is now very long.